Author’s response to reviews

Title: Implementation of health promotion programmes in schools: An approach to understand the influence of contextual factors on the process?

Authors:

Emily Darlington (emily.darlington@univ-lyon1.fr)
Nolwenn Violon (nolwenn.cubizolle@orange.fr)
Didier Jourdan (didier.jourdan@univ-bpclermont.fr)

Version: 1 Date: 13 Sep 2016

Author’s response to reviews:

PUBH-D-16-02714:

"Implementation of health promotion programmes in schools: how do interactions between contextual factors influence the implementation process to result in variable programme impact?"

Dear Editor,

We are pleased to send you our revised manuscript.

The changes requested were as follows:

1. Please add a statement in the manuscript indicating the source of funding for this study.

=> The following statement was added to the manuscript:

“Funding for this work was exclusively provided by local communities. No other source of funding is to declare for this research”
2. Please indicate in the manuscript ethics statement the actual name of the committee that approved the study.

=> The following statement was added to the manuscript:

“The research was approved by the Rectorat Académie de Versailles (Regional Education Authority of Versailles). In France Regional Education Authorities have institutional and legal authority over school level education, higher education level, and research. Research in school settings systematically requires clearance from the Regional Education Authority’s ethical committee to be carried out.”

3. Please also indicate whether the informed consent obtained from the human subjects was written or oral; if oral, please indicate the reason for obtaining oral rather than written consent.

=> The following information were added:

“All participants gave their oral informed consent for inclusion. Oral consent was given during data collection (interviews) and seemed acceptable as the data collected focused on professional issues only, and no issues of personal nature were discussed.”

4. Please add a title page and abstract to the manuscript in accordance with the submission guidelines.

=> Title page and abstract were added to the general manuscript.

We thank you in advance for considering our manuscript for publication

Yours sincerely,

The authors

Emily Darlington,

On behalf of Nolwenn Violon and Didier Jourdan